Author’s response to reviews

Title: Prevalence of obesity and overweight and its associated factors among registered pensioners in Ghana; a cross sectional studies

Authors:

Benjamin Nuertey (ben.nuertey@gmail.com)
Alabira Alhassan (alabira.iddrisu@gmail.com)
Augustine Nuertey (nadtee02@gmail.com)
Isaac Mensah (imensah70@gmail.com)
Victor Adongo (adongo.victor@gmail.com)
Clement Kabutey (kabuteyclement@ymail.com)
Joyce Addai (joyce.nuertey@gmail.com)
Richard Biritwum (biritwum@africaonline.com.gh)

Version: 2 Date: 26 Jun 2017

Author’s response to reviews:

Title: Prevalence of obesity and overweight and its associated factors among registered pensioners in Ghana; a cross sectional studies

Authors:

Benjamin D. Nuertey (ben.nuertey@gmail.com)
Alabira Iddrisu Alhassan (alabira.iddrisu@gmail.com)
Augustine D. Nuertey (nadtee02@gmail.com)
Isaac Asimadu Mensah (imensah70@gmail.com)
Victor Adongo (adongo.victor@gmail.com)
Clement Kabutey (kabuteyclement@ymail.com)
Joyce Addai (joyce.nuertey@gmail.com)
Richard Bekoe Biritwum (briritum@africaonline.com.gh)

Date: 26th June 2017

Dear Editor,

Thank you for assuring us of the potential acceptability of our manuscript for publication in BMC Obesity. This is how we addressed the editors comments.

1. Please provide email addresses to all authors on the front page.

Email addresses of all authors have been provided as suggested

2. Thank you for the ethics approval and consent to participate statement. In this statement please ensure to provide all details of ethics approval: name of the committee that approved the study, details of consent to participate (please clearly state whether that have been oral or written consent). We note that details of ethics approval have been provided elsewhere in the manuscript however we need them to be provided in the Declarations section.

The ethics approval and consent to participate statement have been provided in the declaration section

3. Thank you for the availability of data and materials statement. Unfortunately the statement in the current form is not acceptable. Please state that data and materials are available from the corresponding author. Please include the corresponding author's email address.

The suggestions were carried out

4. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy.

This was equally done

Yours sincerely,

Benjamin D. Nuertey
(ben.nuertey@gmail.com)